Author's response to reviews

Title: Psychological distress as predictor of quality of life in men experiencing infertility: a cross-sectional survey

Authors:

Juliana LR Chachamovich (jurigol@videotron.ca)
Eduardo Chachamovich (echachamovich@videotron.ca)
Hélène Ezer (helene.ezer@videotron.ca)
Fernanda P Cordova (fpcordova@terra.com.br)
Marcelo P Fleck (mfleck.voy@terra.com.br)
Daniela R Kauth (daniela.kauth@gmail.com)
Eduardo P Passos (epp@via-rs.net)

Version: 5 Date: 27 April 2010

Author's response to reviews:

Dear Editorial,

Authors' names have been changed as required.

Sincerely,

Juliana Chachamovich